\begin{table}%ta.1
%\centering
\par
\caption{\label{table.basicdata}Basic data of the Herbig Ae/Be star V1247~Ori.} 
\small  \begin{tabular}{l c l l}
\hline \hline \noalign{\smallskip}
Quantity  			& Value			& Unit		& Reference\tablefootmark{a} \\
            \hline\noalign{\smallskip}
$\alpha^{J2000}$		& 05 38 05.25		&		& TYC	\\
$\delta^{J2000}$		& --01 15 21.7		&		& TYC	\\
$\mu_\alpha \cos{\delta}$	& --0.9 $\pm$ 1.1	& mas~a$^{-1}$	& TYC	\\
$\mu_\delta$			&  +1.3 $\pm$ 1.1	& mas~a$^{-1}$	& TYC	\\
$U$				& 10.29 $\pm$ 0.01:	& mag		& Sh62	\\
$B$				& 10.20 $\pm$ 0.01:	& mag		& Sh62	\\
$B_T$				& 10.18 $\pm$ 0.03	& mag		& TYC	\\
$V$				&  9.88 $\pm$ 0.01:	& mag		& Sh62	\\
$V_T$				&  9.85 $\pm$ 0.03	& mag		& TYC	\\
%$r'$				&  9.771 		& mag		& CMC14	\\
$R$				&  9.68 $\pm$ 0.01:	& mag		& Vi03	\\
$R_C$				&  9.61 $\pm$ 0.05	& mag		& Fu02	\\
%$i'$				&  9.637 $\pm$ 1.0:	& mag		& DENIS	\\
$I$				&  9.44 $\pm$ 0.01:	& mag		& Vi03	\\
$I_C$				&  9.32 $\pm$ 0.05	& mag		& Fu02	\\
$J$				&  8.88 $\pm$ 0.03	& mag		& 2MASS	\\
$H$				&  8.20 $\pm$ 0.05	& mag		& 2MASS	\\
$K_{\rm s}$			&  7.41 $\pm$ 0.03	& mag		& 2MASS	\\
8~$\mu$m			&  0.66 $\pm$ 0.03	& Jy		& MSX6C	\\
12~$\mu$m			&  0.68 $\pm$ 0.07	& Jy		& IPAC	\\
21~$\mu$m			&  2.36 $\pm$ 0.17	& Jy		& MSX6C	\\
25~$\mu$m			&   3.9 $\pm$ 0.3	& Jy		& IPAC	\\
60~$\mu$m			&  10.4 $\pm$ 1.2	& Jy		& IPAC	\\
100~$\mu$m			&   9.0 $\pm$ 0.9	& Jy		& IPAC	\\
Sp. type			& A5~III, F0~Ve		& 		& SC71, Vi03	\\
            \hline
         \end{tabular}
\tablefoot{\tablefoottext{a}{TYC: H{\o}g et~al. (\cite{Hog00});
Sh62: Sharpless (\cite{Shar62}); Vi03: Vieira et~al. (\cite{Vieira03}); Fu02: Fujii et~al. (2002); 2MASS: Skrutskie et~al. (\cite{Skru06}); MSX6C: Egan et~al. (2003) (8 and 21~$\mu$m-band measurements correspond to the
SPIRIT~III/{\em Midcourse Space Experiment} bands~$A$, 8.28~$\mu$m
[6.8--10.8~$\mu$m], and $E$, 21.34~$\mu$m [18.2--25.1~$\mu$m];
IPAC: Joint {IRAS} Science Working Group 1988; SC71: Schild \& Cowley (\cite{Schild71}).}}\vspace{-3mm}
   \end{table}